Author's response to reviews

Title: Exploring new pharmacology and toxicological screening and safety evaluation of one widely used formulation of Nidrakar Bati from South Asia region

Authors:

Afria Zaman (afriabd@gmail.com)
Md Shamsuddin Sultan Khan (jupitex@gmail.com)
Lucky Akter (lucky_akter1986@yahoo.com)
Sharif Hossain Syeed (sharifhs777@gmail.com)
Jakia Akter (pharmasmrity@gmail.com)
Abdullah Al Mamun (mamunal007@yahoo.com)
Md Ershad Alam (ershaduap20@gmail.com)
Md Ahsan Habib (ahsan.bph@gmail.com)
Md Abdul Jalil (jalil_bu15@yahoo.com)

Version: 6
Date: 25 March 2015

Author's response to reviews:

Author response to editorial comments:

Figures, tables and reference list are provided properly. Each figures are uploaded. Tables, figures legends and references list is included in the manuscript document.